Author’s response to reviews

Title: Assessment of the toll-like receptor 4 Asp299gly, thr399Ile and interleukin-8 -251 polymorphisms in the risk for the development of distal gastric cancer

Authors:

Elvira Garza-Gonzalez (elvira_garza_gzz@yahoo.com)
Francisco J Bosques-Padilla (fbosques58@hotmail.com)
Sandra I Mendoza-Ibarra (qfb_sandra_mendoza@yahoo.com)
Juan P Flores-Gutierrez (ujufloresmx@hotmail.com)
Hector J Maldonado-Garza (hectormaldonadog@yahoo.com)
Guillermo I Perez-Perez (perezg02@med.nyu.edu)

Version: 5 Date: 3 April 2007

Author's response to reviews:

April 3rd, 2007

Sandra Le, Ph. D.
Assistant Editor of the BMC Cancer Journal
BMC-series journals
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB
UK

Reference: MS1441771149118416
Dear Dr. Le

Thank you for the review of our manuscript entitled “Assessment of the toll-like receptor 4 Asp299gly, thr399Ile and interleukin-8 -251 polymorphisms in the risk for the development of distal gastric cancer” We have considered the editor’s notes:

Please put all the authors on one line, and ensure that they match with those listed in the submission system exactly, and in an identical order.
Response: The authors were put in one line and they match with those listed in the submission system exactly, and in an identical order.

Please indicate the corresponding author with a symbol.
Response: The corresponding author is now indicated with an asterisk *

Please provide a 'Tables' section heading
Response: A heading was added for the Tables section

Thank you for considering this work.

Sincerely yours,

Guillermo I. Perez Perez, DSc.
Associate Professor of Medicine